Author's response to reviews

Title: The impact of comorbidity and stage on ovarian cancer mortality: A nationwide Danish cohort study

Authors:

Mette S Tetsche (mst@rn.dk)
Claus Dethlefsen (cld@rn.dk)
Lars Pedersen (lap@dce.au.dk)
Henrik T Sorensen (hts@dce.au.dk)
Mette Norgaard (m.noergaard@rn.dk)

Version: 3 Date: 5 January 2008

Author's response to reviews: see over
RE: Manuscript Number: 4588257221616513
‘The impact of comorbidity and stage on ovarian cancer mortality:
A nationwide Danish cohort study’
Corresponding Author: Dr Mette S. Tetsche
Submitted to: BMC Cancer

Thank you very much for giving us the opportunity to revise the manuscript again. We have gone through the constructive comments from the reviewer, and we have revised the manuscript accordingly.

Our response to the comments is as follows:

Reviewer René Scheiden:

1: *Comorbidity and the Charlson Index comorbidity score being a central component of the present study I think that the proposed table (comorbid conditions in the 5,213 ovarian cancer patients) is not too extensive for the paper and must be part of it. For better understanding it is absolutely necessary to include this table in the article.*

**Ad 1.** We have presented the prevalence of the different comorbid diseases included in the Charlson comorbidity index in the paper as requested (Table 1).

We have added “The comorbid conditions in the 5,213 ovarian cancer patients are shown in Table 1.” Page 7 line 2

3: *At the beginning of the rephrased sentence at page 5 line 17 the word â##Forâ## is missing.*

**Ad 3.** We have corrected the sentence.

Yours sincerely,

Mette Skytte Tetsche